Author’s response to reviews

Title: Full-term abdominal extrauterine pregnancy complicated by post-operative ascites with successful outcome: a case report

Authors:

Gwinyai Masukume (parturitions@gmail.com)
Elton Sengurayi (esengus@yahoo.com)
Alfred Muchara (atm4ic@gmail.com)
Emmanuel Mucheni (emucheni@yahoo.com)
Wedu Ndebele (wndebele@yahoo.co.uk)
Solwayo Ngwenya (drsolngwe@yahoo.co.uk)

Version: 5 Date: 17 August 2012

Author’s response to reviews:

Dear JMCR,

We make the following changes in an attempt to make the manuscript clearer:

1) Explained BMI (Body Mass Index) – i.e. the weight in kilograms divided by the square of the height in meters
2) Explained Apgar score and gave the maximum and minimum scores possible
3) Changed +++ to 3+ = 3 g/L (with reference to urine dipstick)
4) Made more precise characterization of the intravenous fluid used – crystalloid
5) Changed Table 1 heading to simply laboratory data
6) Explained likely mechanism of conjugated hyperbilirubinemia and hypoproteinemia
7) Mentioned that the patient was rhesus D positive
8) Changed primiparity to nulliparity
9) Stated explicitly what we mean by future care
10) HIV to HIV (human immune deficiency virus).

Sincerely yours,

GM, ES, AM, EM, WN & SN